Author's response to reviews

Title: The value of INnovative ICT guided disease management combined with Telemonitoring in OUtpatient clinics for Chronic Heart failure patients. Design and methodology of the IN TOUCH study: a multicenter randomised trial

Authors:

Arjen de Vries (a.e.de.vries@thorax.umcg.nl)
Richard de Jong (Richard.deJong@WZA.NL)
Martje van der Wal (m.h.l.van.der.wal@thorax.umcg.nl)
Tiny Jaarsma (tiny.jaarsma@liu.se)
Rene van Dijk (Rene@cavari.nl)
Hans Hillege (h.hillege@tcc.umcg.nl)

Version: 2 Date: 5 July 2011

Author's response to reviews:

1: We have included a abstract in the manuscript

2: We have included a ethics statement and name of the committee that granted the study ethical approval

3: We have included a authors contribution section